Author's response to reviews

Title: Biepicondylar fracture presenting with dislocation of the elbow: a case report

Authors:

Savas Guner (gunersavas@gmail.com)
Sukriye Ilkay Guner (ilkay.guner@hotmail.com)
Mehmet Fethi Ceylan (mfethiceylan@hotmail.com)
Gokay G{"o}rmeli (ggormeli@yahoo.com)
Cemile Ayse Gormeli (ayseyazici@yahoo.com)
Haci Onder (dr_bethesna@hotmail.com)

Version: 9 Date: 21 June 2012

Author's response to reviews: see over
Dear Editor,

We are extremely grateful to you for your valuable recommendations. We have revised our manuscript according to your suggestions. The changes are as mentioned below.

Comment: Please change your title to: 'Biepicondylar fracture presenting with dislocation of the elbow: a case report'
Respond: The title was changed as “Biepicondylar fracture presenting with dislocation of the elbow: a case report”

Comment: Please refer to your paper as a 'case report' as instead of a 'study'.
Respond: We have changed “In this study we evaluated results of elbow biepicondylar fracture with an associated dislocation which was surgically treated by internal fixation in this case.” sentence as “This case report, evaluates the results of the surgical internal fixation of a biepicondylar fracture of the elbow with an associated dislocation.” in introduction section.

Comment: May we request that your manuscript be checked over by a native English speaker and the language revised.
Respond: The manuscript has revised by native English speaker. We have just submitted the journal online system to clear revised copy.

Sincerely yours,

Savas Guner